Author's response to reviews

Title: The oxytocin/vasopressin receptor antagonist atosiban delays the gastric emptying of a semisolid meal compared to saline in human

Authors:

Bodil Ohlsson (bodil.ohlsson@med.lu.se)
Ola Bjorgell (ola.bjorgell@med.lu.se)
Olle Ekberg (olle.ekberg@med.lu.se)
Gassan Darwiche (gassandarwiche@yahoo.com)

Version: 6 Date: 13 March 2006

Author’s response to reviews: see over
Dear Editor,

1. We have read through the manuscript several times and corrected some small changes. Thus, it is proof read.
2. We have checked the title, it is correct in the submission system.
3. The author list is corrected.
4. The short title is deleted on the first page
5. Abstract is changed from Background/Aim till bara Background
6. The underlining in the abstract is deleted
7. The key words are deleted from the manuscript
8. A single competing interest is left now.
9. Acknowledgement is written as a section head now.
10. We have tried to crop the figures more, but we can’t get the figures more close to the edges than what they are now. If You feel it is not enough, You have to contact us again and we have to try once again. However, we can’t make it any better so far.

If there still are things that is needed to be changed, please contact us again. Now We hope this will be published in BMC gastroenterology
Sincerely Bodil Ohlsson